Author's response to reviews

Title: Hepatoprotective effect of curcumin and alpha-tocopherol against cisplatin-induced oxidative stress

Authors:

Sarawoot Palipoch (spalipoch@hotmail.com)
Chuchard Punsawad (Chuchard.pu@wu.ac.th)
Phanit Koomhin (Phanit.ko@wu.ac.th)
Prasit Suwannalert (prasit.sw@mahidol.ac.th)

Version: 5
Date: 7 March 2014

Author's response to reviews: see over
A cover letter giving a point-by-point response to the concerns

Reviewer's report 1 (OK)

Title: Hepatoprotective effect of curcumin and alpha-tocopherol against cisplatin-induced oxidative stress

Version: 4 Date: 21 February 2014

Reviewer: Somchai Pinlaor

Reviewer's report:
The authors have improved ms writing according to the reviewer's comments. Level of interest: An article whose findings are important to those with closely related research interests

Quality of written English: Acceptable

Statistical review: No, the manuscript does not need to be seen by a statistician.

Declaration of competing interests:
I declare that I have no competing interests
Reviewer's report 2 (OK)

Title: Hepatoprotective effect of curcumin and alpha-tocopherol against cisplatin-induced oxidative stress

Version: 4 Date: 26 February 2014

Reviewer: P. S. Bedi

Reviewer's report:

Research article

Hepatoprotective effect of curcumin and alpha-tocopherol against cisplatin-induced oxidative stress

Sarawoot Palipoch, Chuchard Punsawad, Phanit Koomhin and Prasit Suwannalert

BMC Complementary and Alternative Medicine.

Major Compulsory Revisions:
NIL

Minor Essential Revisions:
Nil

Discretionary Revisions:
Nil

Level of Interest:
An article whose findings are important to those with closely related research interests.

Quality of English:
Acceptable

Statistical Review:
Not required

Declaration of competing interests:
Submitted earlier

Declaration of competing interests:
i declare that I have no competing interests
Reviewer's report 3
Title: Hepatoprotective effect of curcumin and alpha-tocopherol against cisplatin-induced oxidative stress
Version: 4 Date: 26 February 2014
Reviewer: George Asare
Reviewer's report:
Abstract. line 41. Change ...reduce to....reduction
  • Done
line 42. change saline group to ...the saline group
  • Done
line 73. change ...causes ....to....cause
  • Done
Level of interest: An article whose findings are important to those with closely related research interests
Quality of written English: Acceptable
Statistical review: No, the manuscript does not need to be seen by a statistician.
Declaration of competing interests:
I declare that I have no competing interest.
Reviewer's report
Title: Hepatoprotective effect of curcumin and alpha-tocopherol against cisplatin-induced oxidative stress
Version: 4 Date: 25 February 2014
Reviewer: Jithan Aukunuru
Reviewer's report:
The manuscript is acceptable in the present form
Level of interest: An article of importance in its field
Quality of written English: Acceptable
Statistical review: Yes, but I do not feel adequately qualified to assess the statistics.
  • OK
Declaration of competing interests:
I don't have any competing interests